Author’s response to reviews

Title: From digital world to real life: a robotic approach to the esophagogastric junction with a 3D printed model

Authors:
Luigi Marano (luigi.marano@unisi.it)
Alessandro Ricci (alessandro.ricci@edific.it)
Vinno Savelli (savelli@unisi.it)
Luigi Verre (verre@unisi.it)
Luca Di Renzo (luca.direnzo@edific.it)
Elia Biccardi (elia.biccardi@edific.it)
Giacomo Costantini (giacomo.costantini@edific.it)
Daniele Marrelli (marrelli@unisi.it)
Franco Roviello (franco.roviello@unisi.it)

Version: 3 Date: 27 Sep 2019

Author’s response to reviews:

We really thank you the Assistant Editor for the comments to improve the quality of the paper. As requested, we properly performed all corrections.

1. Overlap
We really apologize for the overlap that exists throughout our manuscript. Of course we wish to express some of the same ideas contained in other publications (mainly by the Marano et al), however we rephrased all the indicated sentences to minimize the overlap.

2. Ethics
Done

3. Availability of Data and Materials
Done

4. Consent for Publication
Done

5. Corresponding Author Email
Done
6. Image Attribution
We declare that the map depicted in figure 1e and 2b are original and are not taken from another source.

7. Authors Contributions
Done

8. Conclusion
Done

9. Abbreviations
Done

10. Formatting
Done